Author's response to reviews

Title: Cardiac damage after treatment of childhood cancer: A long-term follow-up

Authors:

Veronika Velensek (veronika.velensek@kclj.si)
Uros Mazic (uros.mazic@mf.uni-lj.si)
Ciril Krzisnik (ciril.krzisnik@mf.uni-lj.si)
Damjan Demsar (damjan.demsar@ijs.si)
Janez Jazbec (janez.jazbec@mf.uni-lj.si)
Berta Jereb (bjereb@onko-i.si)

Version: 4 Date: 18 March 2008

Author's response to reviews:

Dear editor
Thank you for your re-revision of our manuscript. We are very happy with the comments of the revisor.
Regarding the additional comments we have added:
1. A paragraph clarifining the ethical aspect of our study was added in the methodology section
2. A paragraph in which we explain the contribution of each author in added at the end of the manuscript just before the references section.
We sincerely hope that the revised version is improved and you will find it suitable for publication in your journal.
Sincerely
Veronika Velensek